Perceived Neurotype of the Other May Affect Self/Other-Representation in Autistic People

Neurodiversity Volume 3: 1–17 © The Author(s) 2025 Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/27546330251339560 journals.sagepub.com/home/ndy



Rachel Moseley¹, Kalai Hung² and Jie Sui³

Abstract

Recent work suggests that the 'social deficits' historically ascribed to autism may be the product of cross-neurotype sociocommunicative differences between autistic people and the neurotypical majority. Where previous work has explored impacts of neurotype mismatches on more complex social behaviour, we aimed to explore how perception of another person's neurotype affects unconscious, implicit processes of self/other-representation that underpin higher-order sociocognitive processes. Autistic (n = 149) and non-autistic (n = 166) participants completed a perceptual matching task where they affirmed or negated learned associations between geometric shapes and three person-labels (themselves, a named friend, and a stranger). The majority of autistic participants perceived their friends and strangers as neurotypical, and the usual preferential processing of friends over strangers was reduced in this group. Effects of other neurotypes were evident in slightly lower accuracy when processing information about people with a different neurodivergent/neurotypical status to participants. The real-life relevance of cognitive biases was indicated by an indirect relationship of greater self-bias to more intense past-year suicide ideation via the mediator of lower autistic community connectedness. Being a neurominority affects implicit processing of socially-relevant information as well as explicit social processes, and these differences may be quantified by a simple cognitive measure linked to complex social behaviour.

Keywords

self-representation, neurotype, double empathy, social cognition

Received: November 2, 2024; accepted: April 17, 2025

Lay Abstract

What is Already Known About this Topic?

Autistic people often have negative social encounters with non-autistic people. Historically, this has been blamed on 'social deficits' in autistic people. We now know that people of different neurotypes communicate differently, and difficulties arise when people of different neurotypes interact. While these effects of neurotype mismatches between individuals are observable in complex social behaviour, we do not know if they are also evident in more fundamental processes underpinning how we think about and experience affiliation with other people.

What Does this Paper Add?

We wanted to see if a perceived neurotype affects the way that people process information about familiar and unfamiliar others (friends vs. strangers). Usually, people are better at processing information about friends than strangers. This is because friends are socially relevant and hence become affiliated to our self-concept, an affiliation which seems to be important for feelings of connectedness. If effects of the other person's neurotype are evident in the way we process information unconsciously, it may go some way to explaining why autistic people often feel socially disconnected from others.

¹Department of Psychology, Bournemouth University, Poole, UK ²Department of Psychology, Tsinghua University, Beijing, China ³School of Psychology, University of Aberdeen, Aberdeen, UK

Corresponding Author:

Rachel Moseley, Department of Psychology, Bournemouth University, Poole, UK. Email: rmoseley@bournemouth.ac.uk

Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access page (https://us. sagepub.com/en-us/nam/open-access-at-sage). Autistic and non-autistic participants in our study completed an online task where they had to quickly process information about themselves, a real-life friend who they identified as neurodivergent or neurotypical, or a stranger. We asked them whether they perceived the stranger as neurotypical or neurodivergent. We also controlled for variables that might affect the results.

We found that participants were poorer when processing information about people with different neurotypes. Most autistic participants perceived their friends and strangers as neurotypical, and they showed no processing advantages for friends over strangers. This may mean that autistic people implicitly feel a greater distance between themselves and neurotypical friends as well as strangers. Our study also suggested that differences in biases towards self- and friends may be related to real-life mental health and social outcomes, although these relationships are complex and need further study.

Implications for Practice, Research and/or Policy

The findings suggest that the perceived neurotype of other people may affect the unconscious way we process information about others, as well as complex, conscious social behaviour. They also suggest that it may be more difficult to process information about people of a different neurotype, possibly because of reduced affiliation with them. These findings may help explain why communication between autistic and non-autistic people can be difficult, and could also be applicable to other neurominority groups, like ADHDers. Unconscious biases in the way we process information can be manipulated, and this may be helpful if these unconscious biases really influence the way people feel.

Recent years have seen the recasting of 'deficits in social communication and social interaction' (American Psychiatric, 2013) as the product of interactional dynamics between neurotypical and neurodivergent minds (Davis & Crompton, 2021; Milton, 2012). Research indicates that differing sociocommunicative behaviour affects friendship, rapport development, conversational ease and problemresolution between autistic and non-autistic people (Chen et al., 2022; Crompton et al., 2020; Funawatari et al., 2024; Geelhand et al., 2024; Heasman & Gillespie, 2018, 2019; Morrison et al., 2020; Rifai et al., 2022; Watts et al., 2024). This literature suggests that non-autistic people are perturbed by the different verbal and non-verbal social behaviour of autistic people (Brewer et al., 2016; Casartelli et al., 2020; Edey et al., 2016), and that this may contribute to prejudice against autistic people (Clin & Kissine, 2023; Stagg et al., 2023). This is important because negative interactions with non-autistic people are a regular source of distress for autistic people (Feller et al., 2023; Parenteau et al., 2023). These findings challenge the traditional notion of sociocognitive 'deficits'

ascribed to autistic people, suggesting instead that observed difficulties could be impacted by cross-neurotype dynamics.

One fundamental yet implicit process, central to social cognition and information processing about other people, is our self-representation. Research has demonstrated that the self-representation is multifaceted: it includes not only higher-level representations, drawing on autobiographical memories and conscious reflections and narratives on own personality, beliefs, identity and worth, but also a more fundamental, biological self, incorporating the distinction between self vs. other and relevant vs. not relevant to me. This more fundamental, biological self, acting as an evolutionarily advantageous 'centre of gravity' (Sui & Gu, 2017) or 'anchoring point' (Scheller & Sui, 2022), implicitly influences information processing by drawing our finite attentional and computational resources towards self-relevant information, which can begin at the early, automatic stages of attentional selection at the gateway to perception (Scheller et al., 2024). A manifestation of this self is a self-prioritisation effect occurring in processing and mnemonic advantages for self-relevant information (Alexopoulos et al., 2012; Cunningham et al., 2008; Desebrock et al., 2022; Sui et al., 2012; Symons & Johnson, 1997), as well as in perceived and preferred distances between self and potentially threatening strangers (Desebrock et al., 2022; Moseley et al., 2022; Sui & Humphreys, 2015). Because many self-related stimuli, like our faces and names, are benefitted by greater familiarity to the individual, associative learning tasks have been a particularly helpful means of delineating prioritisation of the self-concept irrespective of confounds like familiarity. Perceptual matching tasks, for instance, show that when initially arbitrary stimuli, such as shapes, are linked with the self-concept through associative learning during the instruction phase, these stimuli are rapidly 'tagged' as socially relevant through association with the self, and hence lead to more efficient processing, reflected in faster and more accurate responses to shapes associated with the self in the matching task (Scheller & Sui, 2022; Sui et al., 2012, 2023). In a similar vein, their association with the self-concept is believed to explain why information related to familiar others or in-group members is tagged as socially relevant and hence processed more quickly and accurately than that of strangers (Enock et al., 2020; Golubickis et al., 2019; Moradi et al., 2020; Moseley et al., 2022; Sui & Humphreys, 2017). This fundamental involvement of the self-concept, even when processing information about other people, is evinced by the evidence of shared neural substrates for mentalising (understanding the mental states of others) and selfrepresentation (Sui & Gu, 2017; Yankouskaya & Sui, 2022).

The way that people consciously and unconsciously represent their self-concept has important relevance to real life. Differences or 'disturbances' in the self-concept have been identified as core and maintaining factors in various forms of psychopathology such as depression (Giommi et al., 2023; Kyrios et al., 2015), and specific signatures of self-appraisal over time differentiate people with and without the lifetime history of suicidality (Sokol et al., 2024). Importantly, overt or conscious differences in the 'higher-order' self-concept and its relation to others also appear to be reflected in the way people *implicitly* prioritise information about themselves and others: for instance, implicit perception of the distance between the self-concept and familiar and unfamiliar others appears to be linked to conscious feelings of loneliness (Moseley et al., 2022). Relatedly, implicit differences in self- and friendprioritisation can be highly precise indicators of psychopathological states like depression when assessed through non-verbal tasks (Hobbs, Sui et al., 2023; Liu et al., 2022; Sui et al., 2021), and can similarly predict early positive response to anxiolytics (Hobbs et al., 2020; Hobbs, Beck et al., 2023). Beyond reflecting an individual's present state, studies have also demonstrated the potential to harness selfprioritisation biases for health and wellbeing (Caldwell et al., 2018; Verplanken & Sui, 2019), highlighting that there may be clinical and predictive utility in research about implicit self/other processing.

As of yet, the way that autistic people implicitly represent their selves in relation to other people is poorly understood, although self-representation has long been a topic of interest in autism research. Differences in the selfconcept are implied in early development (Lombardo & Baron-Cohen, 2011), and the way autistic adults monitor and differentiate between self- and other-mental states (Burrows et al., 2016; Lai et al., 2019), but investigations of self-prioritisation in autistic people have generated mixed findings. Some studies indicate typical behavioural selfprioritisation in mnemonic and perceptual paradigms (Amodeo et al., 2024; Lind et al., 2020; Williams et al., 2018), and typical attentional capture for own names (Nijhof et al., 2022; Oomen et al., 2022). However, others report reduced self-prioritisation in memory (Burrows et al., 2017; Henderson et al., 2009; Lombardo et al., 2007); reduced attentional capture by own face stimuli (Cygan et al., 2022); reduced prioritisation of self-owned items and self-generated actions (Grisdale et al., 2014; Hartley & Bird, 2023; Wuyun et al., 2020); and diminished self-bias at later stages of perceptual processing (Amodeo et al., 2024).

Where previous investigations of the self-prioritisation effect in autistic people have utilised perceptual matching tasks, they have typically focused on trials where the selfconcept *matches* the shape linked with it through associative learning, where self-prioritisation effects comparable to those seen in the general population were reflected in faster and more accurate affirmation that the self-concept belongs with the shape associated with it (Moseley et al., 2022; Williams et al., 2018). However, mismatching trials, where individuals must negate the pairing of the self-label with a shape not associated with it, are also highly informative. These trials represent a relatively higher processing demand, requiring appropriate top-down inhibition of the bottom-up attentional systems which orient to self-related information, despite it being presented with a mismatched stimulus (Humphreys & Sui, 2016). Self- and friendprioritisation, for these items, manifest in the way that nonautistic people often struggle to 'unhook' themselves from salient self- and friend-concepts to indicate their pairing with the incorrect label. We previously observed that autistic participants were less captured by self- or friend-related items in mismatching trials, suggesting possibly reduced specificity between these concepts and reduced friendprioritisation (Moseley et al., 2022). We also found reduced friend-prioritisation in matching trials, where autistic participants were no more accurate for friends than strangers, and a higher-level processing task where, asked to physically represent the 'emotional' closeness of self-, friendand stranger-items, autistic participants placed friends significantly closer to strangers than did non-autistic people, who tended to keep themselves and their friends away from the socially threatening stranger. A greater perceived distance between self and friend, and a smaller distance between the friend and stranger (i.e., reduced friendprioritisation), were associated with greater loneliness.

This reduced likelihood of associating or 'anchoring' familiar others with the self-concept, in autistic people, might be connected with other observations of reduced identification with in-groups (Skorich & Haslam, 2022), a finding linked to poorer mental health in this population (Maitland et al., 2021). The observation that autistic people were seen to differentiate less between friends and strangers (Moseley et al., 2022) may suggest that both are conceptualised as out-groups to the autistic self. This would indeed appear consistent with the pervasive non-belonging that autistic people experience as marginalised minorities among neurotypical people (Botha et al., 2022; Jones et al., 2022; Zhuang et al., 2023), and, in contrast, the sense of belonging that is associated with identifying with other autistic or neurodivergent people more broadly (Botha et al., 2022; Davies et al., 2024; Watts et al., 2024). The neurotype of other people has not yet been investigated as a potential determinant of how the self is represented in relation to neurotypical and neurodivergent others, but it seems likely to affect these processes given the impact of the other-neurotype on the way autistic people respond to those around them, and the different ways autistic people monitor and present their selves when camouflaging to blend in with non-autistic people (Ai et al., 2022). If both familiar and unfamiliar others feel distant due to their neurotype in relation to autistic individuals, this could explain previous observations of reduced friend-prioritisation as manifest in reduced differentiation between friend and

stranger (Moseley et al., 2022). In this instance, we might expect to see interacting effects of perceived neurotype and diagnosis, with autistic participants showing reduced differentiation between friend and stranger (e.g., reduced friend-prioritisation) if they perceive both to be neurotypical.

Hoping to replicate and lend additional explanatory power to our previous findings of reduced friendprioritisation (Moseley et al., 2022), we investigated the effects of perceived neurotype of the other on implicit information processing about self and others in autistic people. As there is suggestion that autistic people can feel belonging amongst individuals with broader forms of neurodivergence who are equally 'different' (Botha et al., 2022), we examined perceptions of the other as either neurotypical or neurodivergent, conceptualising these as out-group and in-group members to autistic people respectively. We hypothesised an interaction between diagnosis and shapelabel, reflecting reduced differentiation between friend and stranger items in autistic people (Moselev et al., 2022). Incorporating a naturalistic design, we did not expect uniformity in how each group perceived familiar and unfamiliar others, but nonetheless hypothesised that a greater number of autistic individuals would be expected to perceive friends and strangers as neurotypical than nonautistic people to perceive others as neurodivergent. Analysis would be determined by sample size, but if feasible to model the perceived neurotype of others as factors, we expected that those who perceived the other as an outgroup member in relation to their neurotype (e.g., perceiving the other as neurotypical in the case of autistic participants) would differentiate less between the friend and the stranger. In matching trials, this would be reflected in less preferential processing of friends over strangers (i.e., less difference in accuracy between friend/stranger items); in mismatching trials, this would look like greater/easier disengagement from friend stimuli which, through association with the self, would normally be preferentially processed over strangers (e.g., improved ability to 'unhook' from friend items and hence greater accuracy for these items, where normally these would be harder to negate).

As per previous analyses where differences in selfprocessing were examined at the group level and as continuous variables differing across individuals (Lind et al., 2020; Moseley et al., 2022; Williams et al., 2018), we subsequently explored the relevance of these biases to indices of real-world significance for autistic people, following the approach adopted by these previous studies to calculate continuous metrics of self- and friend-bias at the individual level. Since perceived neurotype might affect self- and other-referential processing, we first examined relationships between self-prioritisation, friend-prioritisation and autistic community connectedness, a suggested buffer for the deleterious impact of being a marginalised neurotype (Cage et al., 2022). Since differences in conscious and implicit self-concept have been repeatedly observed in relation to depression and suicidality (Liu et al., 2022; Sokol et al., 2024; Sui et al., 2021), both of which are more prevalent in autistic people (Hedley et al., 2022; Lai, 2023), we secondly examined associations between self- and friend-prioritisation and these outcomes in autistic people.

Methods

Participants

We recruited formally-diagnosed autistic participants (n = 170) living in the United Kingdom or United States through Prolific (a participant recruitment website) between March and July 2022. In order to ensure data quality, we excluded 20 autistic participants whose accuracy on our principal task was lower than 55% (see Analysis). As a result of our exclusions, the final autistic sample (n = 149) comprised 66 cisgender men, 60 cisgender women, and 23 participants who were transgender men, transgender women, or had other transgender and/or non-binary identities. Most were diagnosed as adults (average age 19.7 [*SD*: 9.04], although 19.5% did not report the year of diagnosis).

Between April and May 2022, we used Prolific to recruit age- and sex-matched cisgender non-autistic participants (n = 185) living in the United Kingdom or United States. Most completed the study all at once, but 39.2% were part of a larger study who completed the perceptual matching task approximately 4 weeks before completing the other measures specific to this study. We excluded 19 non-autistic participants on the basis of low task accuracy. The final non-autistic sample (n = 166) comprised 82 men, 79 women, and 5 participants with transgender or non-binary identities. Demographic information for all participants is presented in Table 1.

Procedure and Materials

We received ethical approval for the study via the Faculty Ethics Committee at the first author's institution (#39396). The study (approx. 50 min) comprised three experimental tasks (hosted on Inquisit) and then several standardised scales (hosted on Qualtrics); participants could complete the whole study remotely without contact with the researchers. After consenting to participate and providing demographic information, we asked participants to type in the name of a person they would consider one of their closest friends. We then asked whether that friend was of the same sex or gender identity as the participant; how many years they had known them; how close they felt to them (on a scale from 1, 'not familiar at all', to 7, 'highly familiar'); and, finally, whether they thought (or knew) their friend was neurotypical ('not autistic or neurodivergent in any way') or neurodivergent ('autistic or with another

Table I. Participant Demographics.

| | Autistic group $(n = 149)$ | Non-autistic group ($n = 166$) |
|---|----------------------------|----------------------------------|
| Age | x 27.83 (SD: 6.16), 18–39 | x 27.90 (SD: 6.12), 18–39 |
| Racial and/or ethnic groups identified with: | | |
| % White (/White British or American) | 77.3 | 64.1 |
| % Black (Black British/American, or African American) | 6.3 | 4.4 |
| % Mixed ethnicity | 4.5 | 9.4 |
| % Asian | 2.4 | 9.4 |
| % Hispanic or Latino | 2.4 | 1.9 |
| % Prefer not to answer | 7.1 | 10.8 |
| Highest level of education completed: | | |
| % General Certificate of Secondary Education (United Kingdom) | 10.3 | 12.3 |
| % A-Levels/AS-Levels (United Kingdom) | 26.2 | 15 |
| % High school diploma (United States) | 9.3 | 15 |
| % Scholastic Aptitude Tests (United States) | 0 | 5 |
| % Advanced Placement Tests (United States) | 0 | 17.3 |
| % Technical or community college | 11.7 | 4.4 |
| % Undergraduate/bachelor's degree | 30.1 | 19.7 |
| % Graduate degree (MA, MSc, MPhil, other) | 7.6 | 10 |
| % Doctorate degree (PhD, other) | 4.8 | 1.3 |
| % Do not know/not applicable | 0 | 0 |
| Current employment/work status: | | |
| % Employed part-time | 17.5 | 12.5 |
| % Employed full-time | 34.8 | 45.9 |
| % Self-employed | 9.4 | 5.7 |
| % Student | 17.6 | 16.2 |
| % Looking for work or unable to work | 18.3 | 6.9 |
| % Carer | 2.4 | 1.3 |
| % Retired | 0 | 0.6 |
| % Rather not say/did not answer | 0 | 10.9 |

form of neurodivergence, like ADHD'). We next asked participants to choose a name for an imagined stranger from a list of traditionally male- and female-names, asking them to select a name suggestive that the stranger was of the same sex and gender as themselves.

Using a well-established paradigm (Sui et al., 2012), our experiment then took participants through an associative learning phase where they learnt associations between three geometric shapes and three person-labels (see Figure 1, Part A): 'yourself', 'a friend (who you identified as "[entered name]")', and 'a stranger (who you gave the name "[chosen name]")'. Participants were asked to focus on whether the word matched with the shape. Participants then completed 6 self-paced trials where they were shown examples of 3 matching and 3 mismatching trials and told to indicate whether the word matched with the shape, pressing 'b' to indicate matches and 'v' to indicate mismatches. This was followed by 12 self-paced practice trials, where participants were given feedback after each trial and then overall accuracy and time taken were calculated, and then 12 practice trials at real speed with the same kind of feedback was provided. Consequently, participants completed 3 experimental blocks of 72 trials each, where they were asked to indicate, 'as quickly and accurately as possible', whether word-shape pairings were congruent with those they had learnt. Within each trial (Figure 1, Part B), a fixation cross was shown for 500 ms, followed by a shape-label pairing (500 ms); the screen was blank for 3000 ms, waiting for a participant's response, before feedback was displayed for 500 ms ('correct', 'incorrect', 'slow' if they failed to respond). Between each block, our experiment provided participants with a summary of their timings and accuracy so far, a reminder of the associations they had learnt, and reassurance if they were finding the task difficult.

Each block contained matching (36) and mismatching (36) trials. Matching trials were defined as instances where the shape was presented with the learnt person-label, and divided evenly into 12 self-matching, 12 friend-matching, and 12-stranger matching trials. Mismatching trials were those where the shape was presented with the incorrect person-label, and likewise divided into 12 self-mismatching, 12 friend-mismatching and 12 stranger-mismatching trials. Mismatches were moreover divided evenly between the two alternative person-labels.

After the task, we asked participants to indicate whether they had thought about the stranger as neurotypical or neurodivergent. We then asked them to complete two other experimental tasks (the subject of other reports), followed by measures of autistic community connectedness (autistic participants only), depressive symptoms, and suicide ideation. We used 7 items of the ten-item Autistic Community Connectedness scale (Cage et al., 2022), eschewing the subscale on political activism relating to autism, considered less relevant in the present context, in order to reduce participant burden. Each item is scored from 1–6 (range

7-42) where higher scores reflect greater community connectedness; internal consistency was high in our autistic group ($\alpha = .74$), whose average score was 20.6 (*SD*: 6.3). The Patient Health Questionnaire-9 (PHO-9) (Kroenke & Spitzer, 2002), a brief measure of depression (scores ranging 0-27 with higher values reflecting more severe symptoms), showed high internal consistency in our samples $(\alpha = .88 \text{ and } .9 \text{ for autistic and non-autistic participants})$ respectively). A cut-off of 8 is indicative of major depressive disorder: the average scores of the autistic and nonautistic groups were 12.78 (SD: 6.5, Mdn: 13) and 7.90 (SD: 6.1, Mdn: 7) respectively. We measured the frequency of past-year suicide ideation with a single item from the Self-Injurious Thoughts and Behaviours Interview, short form (SITBI) (Nock et al., 2007), scored from 0 (no suicide ideation) to 6 (suicide ideation occurring every day). We also measured the average intensity/duration of suicide ideation, scored from 1 (suicide ideation which lasted less than a minute) to 6 (suicide ideation which persisted for more than 2 days at a time).

Analysis

Initially, to examine the effect of neurotype-alignment in our primary analysis, we first examined how participant groups *perceived* the neurotype of the friend and stranger with two chi-squared tests of independence (where we compared the distribution of autistic and non-autistic participants between those who perceived the friend and stranger to have the same neurotypical/neurodivergent status as themselves, and those who perceived the friend and stranger to have a different neurotypical/neurodivergent status as themselves). We coded perceived neurotype of friends and strangers where 1 represented the same neurotypical/neurodivergent status (i.e., perceiving the other as neurotypical like yourself, or neurodivergent like yourself), and 2 represented different status (i.e., perceiving the other as neurotypical when neurodivergent yourself or vice versa). We discarded from analyses datasets from 3 non-autistic participants who expressed that they were uncertain of their friend's neurotype (an option not selected by any autistic participants); moreover, missing data on perceived neurotype of friends and/or strangers meant that 23 autistic participants and 15 non-autistic participants were not included, resulting in samples of 126 and 148 respectively in analyses using these variables.

In our subsequent group-level analysis of the perceptual matching task, we followed the previous precedent by focusing on accuracy, which is sensitive to differences between groups (Moseley et al., 2022; Sui & Humphreys, 2017), though we provide the same analyses with reaction time data in Supplementary Materials. For 24 autistic and 22 non-autistic participants out of 315 participants in total, there were notable reductions in accuracy compared to the



Figure 1. Perceptual Matching Task. Note. Part A Depicts Shape-Label Associations Learnt by Participants. Part B Depicts the Timing and Sequence of Experimental Trials.

remaining sample (overall accuracy below 55%). We examined data from these participants, and for them only, excluded trials with no response (likely due to task interruption or short lapses in attention) and trials with correct responses but reaction times more than 3 standard deviations from the mean across all conditions. We removed 41.5% of trials across all 46 participants (on average 89 trials per participant), and re-calculated accuracy for each of them. Altogether, the deleted trials comprised 6.1% of the total data.

Our aforementioned preliminary analyses showed that autistic and non-autistic participants were highly unequal in their perception of friend and stranger neurotype. Having insufficient power to model these as factors, we instead modelled them as covariates in mixed-model ANOVAs aimed to address unbalanced sample sizes between groups. In two $2 \times 3 \times 2$ ANOVAs, we modelled within subject factors (Matching [2] and Shape [3]), and Diagnosis [2] as a between-subject factor. In the first ANOVA, we included only depression as a covariate, given its effect on self/other processing (Liu et al., 2022). In the second, otherwise identical analysis, we included a friend and a stranger neurotype as additional covariates. Sphericity was violated for the Shape factor, so we report Huvnh-Feldt values throughout. Significant interactions were explored with two-tailed between/within-subjects post-hoc tests.

Next, to understand the possible relevance of cognitive biases to real-world outcomes in autistic people, we conducted exploratory Pearson correlation analyses in our autistic group alone. As in previous literature in autistic and non-autistic populations (Liu et al., 2025; Moseley et al., 2022; Williams et al., 2018), we operationalised the extent of self-prioritisation at an individual level by subtracting the average accuracy for Self-matching items from the averaged averages of Friend-matching and Stranger-matching items (such that higher scores reflected greater selfprioritisation); the extent of friend-prioritisation was operationalised by subtracting average accuracy for Friend-matching trials from Stranger-matching trials (such that higher scores reflected greater friend-prioritisation). We examined relationships between these variables and autistic community connectedness, depression, and the frequency and intensity of past-year suicide ideation.

Unfortunately, due to time pressures, the study was not preregistered. We did not request permission from participants to make the data freely available, but it is available on reasonable request. Finally, although the research team included neurotypical and neurodivergent people, we regret that autistic people were not otherwise involved in study design or implementation.

Results

Perceived Neurotype of Friend and Stranger Characters

Autistic and non-autistic participants differed in their perception of the neurotypes of 'friend', $\chi^2(2) = 71.83$, p < .001, and 'stranger', $\chi^2(2) = 258.35$, p < .001, characters. The stranger was perceived as neurotypical by 144 out of 149 autistic people, vs. 143 of 148 non-autistic participants. For the friend, similarly, only 52 of 126 autistic people perceived their friend as neurodivergent like themselves, while 129 of 148 non-autistic participants perceived their friend as neurotypical like themselves. Groups did not differ significantly in the number of years they had known their friend (p = .906), how familiar they felt with them

(p = .600), and the likelihood that their friend shared their sex or gender (p = .09).

Since an insufficient sample size precluded our modelling perceived friend and stranger neurotype as betweensubject factors, we examined their effect through modelling them as covariates in one of two otherwise identical $3 \times 2 \times$ 2 mixed ANOVAs. We first present analyses without controlling for the perceived neurotype of the other.

Self- and Friend-Bias Irrespective of the Perceived Neurotype

With depression as the only covariate, we observed a main effect of Shape, F(1.91, 567.24) = 5.50, p = .005, partial $\eta^2 = .02$, reflecting the greatest accuracy for self, then friend, then stranger shapes. An interaction between Matching × Shape, F(1.98, 586.55) = 13.70, p < .001, partial $\eta^2 = .04$, reflected that this Shape effect was significant in matching trials where the shape was presented with the learned person-label, F(1.85, 547.96) = 13.70, p < .001, partial $\eta^2 = .04$, but not in mismatching trials where shapes were presented with person-labels that were contrary to learned associations (p = .280). There were no other main effects or interactions, including relating to Diagnosis.

Effects of Perceived Neurotype on Self- and Friend-Bias

With the addition of perceived friend and stranger neurotype as covariates, main effects of Diagnosis, F(1, 269) = 5.75, p = .017, partial $\eta^2 = .02$, and the covariate Stranger neurotype, F(1, 269) = 7.83, p = .006, partial $\eta^2 = .03$, reflected that, respectively, autistic participants were more accurate than non-autistic participants; and participants who perceived the stranger as the same broad neurotype as themselves (both neurotypical/neurodivergent) were more accurate.

While differential effects of Diagnosis on self- and friendbias (as reflected Diagnosis × Shape interactions) were absent when not controlling for perceived friend and stranger neurotype, controlling for these covariates revealed a three-way interaction of Matching × Shape × Diagnosis, F(2, 538) =6.85, p = .001, partial $\eta^2 = .03$ (see Figure 2). The effect was driven by a significant interaction of Shape and Diagnosis in mismatching trials, F(2, 538) = 5.89, p = .003, partial $\eta^2 = .02$; the same interaction, in matching trials, was non-significant at p = .084). Between-group analyses revealed that autistic people were significantly more accurate than nonautistic people for mismatching friend, F(1, 270) = 12.95, p <.001, partial $\eta^2 = .05$, and mismatching self-items, F(1, 270)



Figure 2. Accuracy in Matching and Mismatching Trials. Note. Accuracy (% Correct) for Friend, Self and Stranger Items in Matching and Mismatching Trials, Controlling for Depression, Friend and Stranger Neurotype-Alignment. Only the Significant Interaction of Shape × Diagnosis in Mismatching Trials was Explored with Individual Post-hoc Comparisons. Error Bars Reflect 95% Confidence Intervals and Asterisks Reflect Significant Group Differences.

= 10.12, p = .002, partial $\eta^2 = .04$, but not mismatching stranger items (p = .519). Within group comparisons for mismatching trials showed no significant effect of Shape for autistic (p = .104) or non-autistic participants (p = .203). The covariate perceived stranger neurotype also had an effect on the Matching × Shape interaction, F(2, 528) = 7.16, p < .001, partial $\eta^2 = .03$, where, in matching trials, the neurotype of the stranger had the greatest impact on accuracy for stranger items. Caution must be taken since perceived neurotype of friend and stranger were covariates rather than factors, with small samples: however, the data suggested having the same neurotypical or neurodivergent status as the stranger was associated with more accurate performance than having a different status to the stranger. In mismatching trials, the stranger neurotype had the greatest impact on self-items, where those who perceived the stranger's neurotype as accuracy.

Individual Differences in Relationships Between Cognitive Biases and Real-Life Outcomes

Within the autistic group, our individual-level analysis indicated that self-prioritisation and friend-prioritisation were positively correlated (r = .370, p < .001). In relation to outcome measures, weak correlations suggested that greater self-prioritisation (r = -.218, p = .008) was associated with lower autistic community connectedness. While neither self- or friend-prioritisation were associated with mental health-related variables, higher autistic community connectedness was associated with lower intensity of pastyear suicide ideation (r = -.170, p = .038).

Given its relationships to autistic community connectedness, we performed a post-hoc mediation analysis examining the indirect effect of self-prioritisation on intensity of past-year suicide ideation via autistic community connectedness. This analysis, using PROCESS for SPSS (5000 samples, Model 4) (Hayes, 2017), confirmed no direct relationship between the extent of self-prioritisation on suicide ideation (b = -1.08, p = .237), but a significant indirect effect of greater self-prioritisation on suicide ideation via lower autistic community connectedness (b = .46 [bootSE: .29], bootCI: .03, 1.14).

Discussion

Recent work has emphasised that sociocommunicative differences across neurotypes have marked impacts on the ease of communication and rapport between autistic and neurotypical people (Chen et al., 2021; Crompton et al., 2020; Heasman & Gillespie, 2018, 2019; Rifai et al., 2022). This approach contextualises the difficulties autistic people face in the social world as the consequence of being a minority neurotype who are impaired when socialising with neurotypical people, who intuitively 'like' autistic people less (Clin & Kissine, 2023; Stagg et al., 2023). To our knowledge, this is the first study to explore whether the perception of the other as neurotypical or a neurodivergent minority influences more fundamental processes underpinning social cognition. The way people process information about self and others reveals preferential processing of certain information, typically about familiar others and in-groups, which is 'tagged' as socially-relevant to the self (Scheller & Sui, 2022; Sui & Gu, 2017). We explored effects of the other's perceived neurotype on these processes at group level, and explored relationships between real-life outcomes and self- and friend-prioritisation at an individual-level. While our analysis was exploratory and results must be interpreted with caution, several findings were of potential note.

Perceived Neurotype of Another Person Appears to Affect Self/Other-Processing

When we controlled for the markedly different ways that autistic and non-autistic people perceived others' neurotype, we observed differences between diagnostic groups which were previously masked. In neurotypical populations, self-prioritisation effects are most evident in matching trials (Liu et al., 2025). Interestingly, as observed in our previous study (Moselev et al., 2022), differences between autistic and non-autistic people emerged in mismatching trials (reflecting the more complex top-down inhibition of orientation to self-relevant information), rather than in matching trials which primarily reflect bottom-up orienting to selfrelevant information (Humphreys & Sui, 2016; Sui et al., 2024) - suggesting the need to examine this paradigm differently in this group. Another point of replication with our previous work (Moseley et al., 2022) was that autistic people performed significantly better than non-autistic participants when negating mismatching self and friend trials. Direct comparison of the two studies is complicated by the fact that herein we controlled for depression in our sample so as to more accurately examine autistic status, but otherwise, there as here, this reduced 'stuckness' on mismatching trials might reflect reduced attentional capture by self and friend items, and hence reduced self- and friendprioritisation in autistic people.

Our control for depression in this study might explain why the effect of 'autism diagnosis' was stronger in our previous study, where it might have reflected other uncontrolled-for differences between groups. A slightly weaker effect in the present study would also be consistent with the way in which the effect of Diagnosis was unmasked by modelling perceived neurotype as a covariate. While our original and preferred approach had been to treat perceived neurotype as a factor, the imbalance in the data necessitated using it as a covariate, a valid approach to control for perceived neurotype. By accounting for the binary perceived neurotype, we removed variance in accuracy that would otherwise have inflated the error term, especially given the uneven distribution of perceived neurotype in friends and strangers. This reduction in residual variance might have effectively sharpened the estimate of the Diagnosis effect, potentially changing a non-significant result into a significant one with the covariate accounted for.

Unsurprisingly, controlling for perceived neurotype had a negligible effect on non-autistic accuracy for self and friend mismatching trials; unfortunately, since almost all non-autistic people perceived strangers and friends to be neurotypical like themselves, we were unable to look at effects of the other's neurotype on non-autistic participants' performance. However, improved autistic performance on these items, once controlling for perceived neurotype, produced the significant group differences reflecting reduced self- and friend-prioritisation. More broadly, across all participants, perceiving the stranger as having a different neurotypical/neurodivergent status to yourself was associated with poorer performance in stranger-matching trials and self-mismatching trials.

These findings suggest that in addition to affecting more complex social behaviour, the perceived neurotype of the other may affect implicit or unconscious self-/otherprocessing, exerting an additional processing demand which non-autistic people do not typically need to navigate. Why might it be more difficult to process information about people whose neurotypical or neurodivergent status differs from one's own? One possible explanation lies in the bias people show towards positive and rewarding stimuli (Yankouskaya et al., 2023). Real-world social difficulties, which arise as a result of differing sociocommunicative behaviour between neurotypical people and autistic people (Chen et al., 2021; Crompton et al., 2020; Heasman & Gillespie, 2018, 2019; Rifai et al., 2022), and indeed neurotypical people and ADHDers (Godfrey-Harris & Shaw, 2023), might mean that perceiving a familiar or unfamiliar other as neurotypical, when neurodivergent oneself, imbues them with some level of negative association or threat. Pleasurable interactions and feelings of belonging between autistic people, and between autistic and other neurodivergent people (Botha et al., 2022; Crompton et al., 2020; Watts et al., 2024), might similarly create a bias towards people who are perceived as neurodivergent minorities like the individual, who might hence constitute a sociallysalient in-group (Enock et al., 2020; Golubickis et al., 2019; Moradi et al., 2020; Moseley et al., 2022; Sui & Humphreys, 2017). Since neurotypical/neurodivergent status separated many autistic participants from their friend and the stranger, this fundamental difference might explain why autistic people were less 'stuck' on friend-items once we had tried to control for the statistical variance associated with perceived neurotype. This additional separateness might also explain why participants who perceived the stranger as differing in their neurotypical/neurodivergent status found it harder to confirm matches with these items.

While we cannot presently confirm either interpretation, or ascertain cause and effect between poorer processing and reduced identification, we do know that social identification is important for mental health and moderates the beneficial properties of social connections (Steffens et al., 2021). Accordingly, reduced social identification would be consistent with the link previously observed between loneliness and reduced friend/stranger differentiation in autistic people (Moseley et al., 2022), and the beneficial effects we and others observed in relation to autistic and neurodivergent community connectedness (Botha et al., 2022; Cage et al., 2022; Watts et al., 2024). While researchers have suggested deleterious consequences of reduced social identification in autistic people (Skorich & Haslam, 2022), they have situated this as a deficit inherent to autistic people without considering the societal marginalisation of the same (Turnock

et al., 2022). We encourage future research to consider both intra- *and* extrapersonal factors that might give rise to these processing differences, while attempting to connect implicit to explicit effects of neurotype differences on social behaviour and wellbeing.

Self-Prioritisation was Linked to Suicide Ideation Through Lower Community Connectedness

As a preliminary means of exploring possible relationships between cognitive biases and outcomes relevant to autistic people, we explored relationships between self- and friendprioritisation and social connectedness to other autistic people, depression and two indices of past-year suicide ideation. Both heightened self- and friend-prioritisation in matching trials were associated with lower connectedness to the autistic community. However, through this mediator, greater self-prioritisation was also indirectly associated with greater intensity of past-year suicide ideation. This analysis was post-hoc and exploratory, and cannot support directional relationships between variables. It does, however, support the possible relevance of implicit cognitive selfprioritisation in the processes that underpin suicide ideation, as per relationships between self-prioritisation and depression in neurotypical people (Hobbs, Beck et al., 2023; Liu et al., 2022).

The absence of direct relationships between self- and friend-prioritisation and psychopathology might suggest that these relationships operate differently in autistic people, but could equally reflect insufficient power, the ubiquity of psychopathology in autistic people, and the relative simplicity of the task vs. the longitudinal complexity of relationships with health and social outcomes. While the present analysis was exploratory and could not ascertain directional relationships between variables, we encourage further efforts to address the paucity of research in this area, given the clinical applicability of cognitive biases in detecting and treating psychopathology in the general population (Fodor et al., 2020; Hobbs, Beck et al., 2023; Liu et al., 2022; Sui et al., 2021).

Limitations and Future Directions

Given that our study was exploratory and analyses were not corrected for multiple statistical comparisons, it is possible that significant differences between groups arose as false positives (type 1 errors). The results require replication through more rigorously designed investigations. While we decided to observe the effects of other-neurotype as it naturally occurs in participants' perceptions of friends and hypothetical strangers (rather than experimentally, and artificially, manipulating these), the imbalance between autistic and non-autistic participants prevented us from modelling the perceived neurotype of the other as an independent

variable effectively. For the stranger neurotype in particular, it is possible that the effects of this variable were disproportionately affected by comparisons involving small groups (e.g., with 143 non-autistic and 5 autistic people perceiving the stranger to have the same neurotypical/neurodivergent status as themselves, and 144 autistic people and 5 non-autistic people perceiving the stranger to have a different neurotypical/ neurodivergent status). Experimental manipulation of the way participants perceive friends and strangers might help clarify and establish directionality in the effects and relationships observed here. While explicit labelling of others' neurotypes might create bias in participants (Geelhand et al., 2024; Livingston et al., 2024) and we avoided it for that reason, future manipulations might take advantage of the fact that in reality, individuals often do not know the neurotype of those they interact with, responding instead to verbal and nonverbal cues which affect their behaviour (Geelhand et al., 2024; Whelpley & May, 2023). Incorporating more naturalistic elements into this design, such as subtle cuing as to the neurotype of the unfamiliar other, would allow for more ecologically valid investigation of cross-neurotype effects on implicit social processing, as well as addressing the question as to whether differing sociocommunicative behaviour affects self/other representation in the same way as known or perceived neurotypical/neurodivergent status.

We were unable to validate autism diagnoses in our participants, although the autistic sample in question were those who, via Prolific, indicated that they did possess a formal diagnosis by a qualified specialist. Our late-diagnosed, highly educated autistic sample were unrepresentative of the whole autistic community, most notably autistic people with higher support needs which would have precluded their participation. Several marginalised groups, including people of colour, non-binary and/or transgender people, were underrepresented. Recruiting from Prolific meant that we excluded those with lower computer-literacy and without access to this technology, and that our participants might not reflect a wider autistic population not seeking to engage in psychological research. Notably, our non-autistic population, whose nonautistic status was similarly unconfirmed, may have also been atypical, as perhaps indicated by relatively high levels of depression. Previous studies are suggestive of a greater prevalence of mental health conditions in Prolific than in undergraduate samples (Stanton et al., 2022), and so, although our analyses controlled for depression specifically, the nonautistic group may have been unrepresentative of broader population norms (Borodovsky, 2022).

There is a need for further investigation to interrogate the theoretical premises of our work. The majority of empirical research investigating cross-neurotype social dynamics, much of it stemming from the concept of 'double empathy' (Milton, 2012), focuses on interactions and social processes between autistic and non-autistic people (Chen et al., 2022; Crompton et al., 2020; Funawatari et al., 2024; Geelhand et al., 2024; Heasman & Gillespie, 2018, 2019; Morrison

et al., 2020; Rifai et al., 2022; Watts et al., 2024). In our study, our choice to examine the effects of perception along the neurotypical/neurodivergent divide was based on reported feelings of belongingness that autistic people can feel among other neurominorities (Botha et al., 2022), and the shared experiences of the same. Among these neurominorities, ADHDers, in particular, are another highly stigmatised and victimised neurominority group (Aguado-Gracia et al., 2021; Visser et al., 2024), and one which, indeed, also has poorer wellbeing and higher suicide rates (Fuller-Thomson et al., 2022). However, our design disallowed specification as to whether the friend and stranger's neurominority neurotype precisely matched that of our autistic participants. While the concept of double empathy requires more rigorous experimental exploration (Livingston et al., 2024), one area of development pertains to social communication and relations between neurotypical people and those of different neurominority identities, and between individuals of different neurominority identities (e.g., autistic, ADHD, specific learning difficulties). We venture to suggest that effects of other-neurotype on implicit and (downstream) explicit social processes could be expected to pertain to ADHDers, who also show reduced self-prioritisation and distinct neural substrates underpinning self and emotion processing (Ahmed et al., 2024; Rafi et al., 2023) and who also appear to fall foul of mixedneurotype social dynamics (Godfrey-Harris & Shaw, 2023).

Conclusions

While replicating previous findings of reduced 'capture' by self- and friend-related information in mismatching trials, these preliminary results suggest that the perceived neurotype of the other may affect how autistic people implicitly process information about themselves and others, adding computational demands that might reflect the separateness of both familiar and unfamiliar neurotypical others from the autistic individual. We observed relationships between greater selfprioritisation and greater intensity of past-year suicide ideation, mediated by lower autistic community connectedness. While the findings require replication in designs suitable for drawing causal inferences, they suggest the relevance of self-prioritisation to social functioning and wellbeing. Further research should explore the social and clinical relevance of cognitive biases related to self/other-processing in directional, focused and rigorously controlled study designs.

Acknowledgements

The authors extend our deep gratitude to all participants for their time. We further thank our institutions for the funding which made this research possible.

ORCID iDs

Rachel Moseley D https://orcid.org/0000-0002-5985-6175 Kalai Hung D https://orcid.org/0000-0001-7769-2875 Jie Sui D https://orcid.org/0000-0002-4031-4456

Ethical Considerations

The study was reviewed and approved by the Science and Technology Faculty Ethics Panel at Bournemouth University (#39396). It was performed in accordance with the ethical standards set out in the 1964 Helsinki Declaration.

Consent to Participate

Participants were fully informed about the nature and content of the study, and all gave written consent to participate.

Consent for Publication

Participants consented in writing to publication of the anonymous data.

Authors Contributions

The perceptual matching paradigm was designed by JS, whose body of work primarily underpinned the study. Contributions were as follows: conceptualisation and study design (RLM, KH, led by JS); funding acquisition (JS); programming the experiment (JS, KH); collecting the data (RLM, KH); supervising data collection (JS); preparing the data (JS, KH); analysing the data (JS, KH, RLM); visualising the data (JS, KH, RLM); writing the original draft (RLM); writing and refining the manuscript (JS, KH, RLM). All authors read and approved the final manuscript.

Funding

The study received internal institutional funding granted to JS. No other funding was received for this work.

Declaration of Conflicting Interests

The authors declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Data Availability Statement

Participants did not provide consent for data to be hosted in publicly available repositories. However, data and materials are available on reasonable request.

Supplemental Material

Supplemental material for this article is available online.

References

Aguado-Gracia, J., Mundo-Cid, P., Lopez-Seco, F., Acosta-García, S., Cortes-Ruiz, M. J., Vilella, E., & Masana-Marín, A. (2021). Lifetime victimization in children and adolescents with ADHD. *Journal of Interpersonal Violence*, 36(5–6), NP3241–NP3262. https://doi.org/10.1177/ 0886260518771680

- Ahmed, Z., Cunningham, S. J., Rhodes, S., Gow, A., Macmillan, K., Hutchison, J., & Ross, J. (2024). The self-reference effect in attention deficit hyperactivity disorder. *British Journal of Developmental Psychology*, 42(3), 348–358. https://doi.org/ 10.1111/bjdp.12489
- Ai, W., Cunningham, W. A., & Lai, M.-C. (2022). Reconsidering autistic 'camouflaging' as transactional impression management. *Trends in Cognitive Sciences*, 26(8), 631–645. https:// doi.org/10.1016/j.tics.2022.05.002
- Alexopoulos, T., Muller, D., Ric, F., & Marendaz, C. (2012). I, me, mine: Automatic attentional capture by self-related stimuli. *European Journal of Social Psychology*, 42(6), 770–779. https://doi.org/10.1002/ejsp.1882
- American Psychiatric, A. (2013). Diagnostic and statistical manual of mental disorders (5th ed.). American Psychiatric Association.
- Amodeo, L., Goris, J., Nijhof, A. D., & Wiersema, J. R. (2024). Electrophysiological correlates of self-related processing in adults with autism. *Cognitive, Affective, & Behavioral Neuroscience*, 1–17. https://doi.org/10.1007/s10803-024-06705-8
- Borodovsky, J. T. (2022). Generalizability and representativeness: Considerations for internet-based research on substance use behaviors. *Experimental and Clinical Psychopharmacology*, 30(4), 466–477. https://doi.org/10.1037/pha0000581
- Botha, M., Dibb, B., & Frost, D. (2022). "It's being a part of a grand tradition, a grand counter-culture": A qualitative investigation of autistic community connectedness. *Autism*, 26(8), 2151–2164. https://doi.org/10.1177/13623613221080248
- Brewer, R., Biotti, F., Catmur, C., Press, C., Happé, F., Cook, R., & Bird, G. (2016). Can neurotypical individuals read autistic facial expressions? Atypical production of emotional facial expressions in autism spectrum disorders. *Autism Research*, 9(2), 262–271. https://doi.org/10.1002/aur.1508
- Burrows, C. A., Laird, A. R., & Uddin, L. Q. (2016). Functional connectivity of brain regions for self- and other-evaluation in children, adolescents and adults with autism. *Developmental Science*, 19(4), 564–580. https://doi.org/10. 1111/desc.12400
- Burrows, C. A., Usher, L. V., Mundy, P. C., & Henderson, H. A. (2017). The salience of the self: Self-referential processing and internalizing problems in children and adolescents with autism spectrum disorder. *Autism Research*, 10(5), 949–960. https:// doi.org/10.1002/aur.1727
- Cage, E., Cranney, R., & Botha, M. (2022). Brief report: Does autistic community connectedness moderate the relationship between masking and wellbeing? *Autism in Adulthood*, 4(3), 247–253. https://doi.org/10.1089/aut.2021.0096
- Caldwell, A. E., Masters, K. S., Peters, J. C., Bryan, A. D., Grigsby, J., Hooker, S. A., Wyatt, H. R., & Hill, J. O. (2018). Harnessing centred identity transformation to reduce executive function burden for maintenance of health behaviour change: The Maintain IT model. *Health Psychology Review*, *12*(3), 231–253. https://doi.org/10.1080/17437199.2018. 1437551

- Casartelli, L., Federici, A., Fumagalli, L., Cesareo, A., Nicoli, M., Ronconi, L., Vitale, A., Molteni, M., Rizzolatti, G., & Sinigaglia, C. (2020). Neurotypical individuals fail to understand action vitality form in children with autism spectrum disorder. *Proceedings of the National Academy of Sciences*, *117*(44), 27712–27718. https://doi.org/10.1073/pnas. 2011311117
- Chen, Y.-L., Schneider, M., & Patten, K. (2022). Exploring the role of interpersonal contexts in peer relationships among autistic and non-autistic youth in integrated education. *Frontiers in Psychology*, *13*, 946651. https://doi.org/10.3389/fpsyg. 2022.946651
- Chen, Y.-L., Senande, L. L., Thorsen, M., & Patten, K. (2021). Peer preferences and characteristics of same-group and crossgroup social interactions among autistic and non-autistic adolescents. *Autism*, 25(7), 1885–1900. https://doi.org/10.1177/ 13623613211005918
- Clin, E., & Kissine, M. (2023). Neurotypical, but not autistic, adults might experience distress when looking at someone avoiding eye contact: A live face-to-face paradigm. *Autism*, 27(7), 1949–1959. https://doi.org/10.1177/13623613221 148553
- Crompton, C. J., Hallett, S., Ropar, D., Flynn, E., & Fletcher-Watson, S. (2020). 'I never realised everybody felt as happy as I do when I am around autistic people': A thematic analysis of autistic adults' relationships with autistic and neurotypical friends and family. *Autism*, 24(6), 1438–1448. https://doi.org/10.1177/1362361320908976
- Cunningham, S. J., Turk, D. J., Macdonald, L. M., & Neil Macrae, C. (2008). Yours or mine? Ownership and memory. *Consciousness and Cognition*, 17(1), 312–318. https://doi. org/10.1016/j.concog.2007.04.003
- Cygan, H. B., Nowicka, M. M., & Nowicka, A. J. A. R. (2022). Impaired attentional bias toward one's own face in autism spectrum disorder: ERP evidence. *Autism Research*, 15(2), 241–253. https://doi.org/10.1002/aur.2647
- Davies, J., Cooper, K., Killick, E., Sam, E., Healy, M., Thompson, G., Mandy, W., Redmayne, B., & Crane, L. (2024). Autistic identity: A systematic review of quantitative research. *Autism Research*, 17(5), 874–897. https://doi.org/ 10.1002/aur.3105
- Davis, R., & Crompton, C. (2021). What do new findings about social interaction in autistic adults mean for neurodevelopmental research? *Perspectives on Psychological Science*, 16(3), 649–653. https://doi.org/10.1177/1745691620958010
- Desebrock, C., Barutchu, A., & Spence, C. (2022). The influence of empathy and perceived closeness on self-and friend-biases in arm movements. *Journal of Experimental Psychology: Human Perception and Performance*, 48(9), 953–971. https://doi.org/10.1037/xhp0001028
- Edey, R., Cook, J., Brewer, R., Johnson, M. H., Bird, G., & Press, C. (2016). Interaction takes two: Typical adults exhibit mindblindness towards those with autism spectrum disorder. *Journal of Abnormal Psychology*, 125(7), 879–885. https:// doi.org/10.1037/abn0000199

- Enock, F. E., Hewstone, M. R., Lockwood, P. L., & Sui, J. (2020). Overlap in processing advantages for minimal ingroups and the self. *Scientific Reports*, 10(1), 18933. https://doi.org/10.1038/ s41598-020-76001-9
- Feller, C., Ilen, L., Eliez, S., & Schneider, M. (2023). Characterizing daily-life social interactions in adolescents and young adults with neurodevelopmental disorders: A comparison between individuals with Autism Spectrum Disorders and 22q11. 2 deletion syndrome. *Journal of Autism and Developmental Disorders*, 53(1), 245–262. https://doi.org/10. 1007/s10803-021-05423-9
- Fodor, L. A., Georgescu, R., Cuijpers, P., Szamoskozi, Ş, David, D., Furukawa, T. A., & Cristea, I. A. (2020). Efficacy of cognitive bias modification interventions in anxiety and depressive disorders: A systematic review and network meta-analysis. *The Lancet Psychiatry*, 7(6), 506–514. https://doi.org/10.1016/ S2215-0366(20)30130-9
- Fuller-Thomson, E., Rivière, R. N., Carrique, L., & Agbeyaka, S. (2022). The dark side of ADHD: Factors associated with suicide attempts among those with ADHD in a national representative Canadian sample. *Archives of Suicide Research*, 26(3), 1122–1140. https://doi.org/10.1080/13811118.2020. 1856258
- Funawatari, R., Sumiya, M., Iwabuchi, T., Nishimura, T., Komeda, H., & Senju, A. (2024). Camouflaging in autistic adults is modulated by autistic and neurotypical characteristics of interaction partners. *Journal of Autism and Developmental Disorders*, 1–11. https://doi.org/10.1007/ s10803-024-06481-5
- Geelhand, P., Papastamou, F., Jaspard, S., & Kissine, M. (2024). Autistic adults display different verbal behavior only in mixedneurotype interactions: Evidence from a referential communication task. *Autism*, 13623613241298376. https://doi.org/10. 1177/13623613241298376
- Giommi, F., Bauer, P. R., Berkovich-Ohana, A., Barendregt, H., Brown, K. W., Gallagher, S., Nyklíček, I., Ostafin, B., Raffone, A., Slagter, H. A., Trautwein, F.-M., & Vago, D. R. (2023). The (In) flexible self: Psychopathology, mindfulness, and neuroscience. *International Journal of Clinical and Health Psychology*, 23(4), 100381. https://doi.org/10.1016/j. ijchp.2023.100381
- Godfrey-Harris, M., & Shaw, S. C. K. (2023). The experiences of medical students with ADHD: A phenomenological study. *PLoS One*, 18(8), e0290513. https://doi.org/10.1371/journal. pone.0290513
- Golubickis, M., Ho, N. S., Falbén, J. K., Mackenzie, K. M., Boschetti, A., Cunningham, W. A., & Neil Macrae, C. (2019). Mine or mother's? Exploring the self-ownership effect across cultures. *Culture and Brain*, 7, 1–25. https://doi.org/10. 1007/s40167-018-0068-0
- Grisdale, E., Lind, S. E., Eacott, M. J., & Williams, D. M. (2014). Self-referential memory in autism spectrum disorder and typical development: Exploring the ownership effect. *Consciousness and Cognition*, 30, 133–141. https://doi.org/ 10.1016/j.concog.2014.08.023

- Hartley, C., & Bird, L.-A. (2023). Exploring the influence of object similarity and desirability on children's ownership identification and preferences in autism and typical development. *Journal of Autism and Developmental Disorders*, 53, 2362– 2372. https://doi.org/10.1007/s10803-022-05489-z
- Hayes, A. F. (2017). An introduction to mediation, moderation, and conditional process analysis: A regression-based approach (Second Edition). Guilford Press.
- Heasman, B., & Gillespie, A. (2018). Perspective-taking is twosided: Misunderstandings between people with Asperger's syndrome and their family members. *Autism*, 22(6), 740–750. https://doi.org/10.1177/1362361317708287
- Heasman, B., & Gillespie, A. (2019). Neurodivergent intersubjectivity: Distinctive features of how autistic people create shared understanding. *Autism*, 23(4), 910–921. https://doi.org/10. 1177/1362361318785172
- Hedley, D., Hayward, S. H., Clarke, A., Uljarević, M., & Stokes, M. A. (2022). Suicide and autism: A lifespan perspective. In End of Life and People with Intellectual and Developmental Disability (pp. 59–94). Springer.
- Henderson, H. A., Zahka, N. B., Kojkowski, N. M., Inge, A. P., Schwartz, C. B., Hileman, C. M., Coman, D. C., & Mundy, P. C. (2009). Self-referenced memory, social cognition, and symptom presentation in autism. *Journal of Child Psychology and Psychiatry*, 50(7), 853–861. https://doi.org/ 10.1111/j.1469-7610.2008.02059.x
- Hobbs, C., Beck, M., Denham, F., Pettitt, L., Faraway, J., Munafò, M. R., Sui, J., Kessler, D., & Button, K. (2023). Relationship between change in social evaluation learning and mood in early antidepressant treatment: A prospective cohort study in primary care. *Journal of Psychopharmacology*, *37*(3), 303–312. https://doi.org/10. 1177/02698811221116928
- Hobbs, C., Murphy, S. E., Wright, L., Carson, J., Van Assche, I., O'Brien, J., Oyesanya, M., Sui, J., Munafò, M. R., Kessler, D., Harmer, C. J., & Button, K. S. (2020). Effect of acute citalopram on self-referential emotional processing and social cognition in healthy volunteers. *BJPsych Open*, 6(6), e124. https:// doi.org/10.1192/bjo.2020.107
- Hobbs, C., Sui, J., Kessler, D., Munafö, M. R., & Button, K. S. (2023). Self-processing in relation to emotion and reward processing in depression. *Psychological Medicine*, 53(5), 1924– 1936. https://doi.org/10.1017/S0033291721003597
- Humphreys, G. W., & Sui, J. (2016). Attentional control and the self: The Self-Attention Network (SAN). *Cognitive Neuroscience*, 7(1-4), 5–17. https://doi.org/10.1080/ 17588928.2015.1044427
- Jones, S. C., Gordon, C. S., Akram, M., Murphy, N., & Sharkie, F. (2022). Inclusion, exclusion and isolation of autistic people: Community attitudes and autistic people's experiences. *Journal of Autism and Developmental Disorders*, 52(3), 1131–1142. https://doi.org/10.1007/s10803-021-04998-7
- Kroenke, K., & Spitzer, R. L. (2002). The PHQ-9: A new depression diagnostic and severity measure. *Psychiatric Annals*, 32(9), 509– 515. https://doi.org/10.3928/0048-5713-20020901-06

- Kyrios, M., Nelson, B., Ahern, C., Fuchs, T., & Parnas, J. J. P. (2015). The self in psychopathology. *Psychopathology*, 48(5), 275–277. https://doi.org/10.1159/000438876
- Lai, M.-C. (2023). Mental health challenges faced by autistic people. *Nature Human Behaviour*, 7(10), 1620–1637. https://doi. org/10.1038/s41562-023-01718-2
- Lai, M.-C., Lombardo, M. V., Chakrabarti, B., Ruigrok, A. N., Bullmore, E. T., Suckling, J., Auyeung, B., Happé, F., Szatmari, P., Baron-Cohen, S., Bailey, A. J., Bolton, P. F., Carrington, S., Catani, M., Craig, M. C., Daly, E. M., Deoni, S. C. L., Ecker, C., & Henty, J., ... S. C. Williams (2019). Neural self-representation in autistic women and association with 'compensatory camouflaging'. *Autism*, 23(5), 1210–1223. https://doi.org/10.1177/1362361318807159
- Lind, S. E., Williams, D. M., Nicholson, T., Grainger, C., & Carruthers, P. (2020). The self-reference effect on memory is not diminished in autism: Three studies of incidental and explicit self-referential recognition memory in autistic and neurotypical adults and adolescents. *Journal of Abnormal Psychology*, *129*(2), 224–236. https://doi.org/10.1037/abn0000467
- Liu, Z., Hu, M., Zheng, Y., Sui, J., & Chuan-Peng, H. (2025). A multiverse assessment of the reliability of the self-matching task as a measurement of the self-prioritization effect. *Behavior Research Methods*, 57(1), 1–20. https://doi.org/ 10.3758/s13428-024-02538-6
- Liu, Y. S., Song, Y., Lee, N. A., Bennett, D. M., Button, K. S., Greenshaw, A., Cao, B., & Sui, J. (2022). Depression screening using a non-verbal self-association task: A machinelearning based pilot study. *Journal of Affective Disorders*, 310, 87–95. https://doi.org/10.1016/j.jad.2022.04.122
- Livingston, L. A., Hargitai, L. D., & Shah, P. (2025). The double empathy problem: A derivation chain analysis and cautionary note. *Psychological Review*, 132(3), 744–757. https://doi.org/ 10.1037/rev0000468
- Lombardo, M. V., Barnes, J. L., Wheelwright, S. J., & Baron-Cohen, S. (2007). Self-referential cognition and empathy in autism. *PLoS One*, 2(9), e883. https://doi.org/10. 1371/journal.pone.0000883
- Lombardo, M. V., & Baron-Cohen, S. (2011). The role of the self in mindblindness in autism. *Consciousness and Cognition*, 20(1), 130–140. https://doi.org/10.1016/j.concog.2010.09.006.
- Maitland, C. A., Rhodes, S., O'Hare, A., & Stewart, M. E. (2021). Social identities and mental well-being in autistic adults. *Autism*, 25(6), 1771–1783. https://doi.org/10.1177/1362361 3211004328
- Milton, D. E. (2012). On the ontological status of autism: The 'double empathy problem'. *Disability & Society*, 27(6), 883–887. https://doi.org/10.1080/09687599.2012.710008
- Moradi, Z., Najlerahim, A., Macrae, C. N., & Humphreys, G. W. (2020). Attentional saliency and ingroup biases: From society to the brain. *Social Neuroscience*, 15(3), 324–333. https://doi. org/10.1080/17470919.2020.1716070
- Morrison, K. E., DeBrabander, K. M., Jones, D. R., Faso, D. J., Ackerman, R. A., & Sasson, N. J. (2020). Outcomes of realworld social interaction for autistic adults paired with

autistic compared to typically developing partners. *Autism*, 24(5), 1067–1080. https://doi.org/10.1177/136236131989 2701

- Moseley, R., Liu, C. H., Gregory, N., Smith, P., Baron-Cohen, S., & Sui, J. (2022). Levels of self-representation and their sociocognitive correlates in late-diagnosed autistic adults. *Journal of Autism and Developmental Disorders*, 52(7), 3246–3259. https://doi.org/10.1007/s10803-021-05251-x
- Nijhof, A. D., von Trott zu Solz, J., Catmur, C., & Bird, G. (2022). Equivalent own name bias in autism: An EEG study of the Attentional Blink. *Cognitive, Affective, & Behavioral Neuroscience, 22*(3), 625–639. https://doi.org/10.3758/ s13415-021-00967-w
- Nock, M. K., Holmberg, E. B., Photos, V. I., & Michel, B. D. (2007). Self-Injurious thoughts and behaviors interview: Development, reliability, and validity in an adolescent sample. *Psychological Assessment*, 19(3), 309–317. https://doi.org/10. 1037/1040-3590.19.3.309
- Oomen, D., El Kaddouri, R., Brass, M., & Wiersema, J. R. (2022). Neural correlates of own name and own face processing in neurotypical adults scoring low versus high on symptomatology of autism spectrum disorder. *Biological Psychology*, 172, 108358. https://doi.org/10.1016/j.biopsycho.2022.108358
- Parenteau, C. I., Lampinen, L. L., Ghods, S. S., Taylor, J. L., Adams, R. E., Bishop, S. L., & Zheng, S. (2023). Self-reported everyday sources of happiness and unhappiness in autistic adults. *Journal of Autism and Developmental Disorders*, 54(4), 1538–1548. https://doi.org/10.1007/s10803-023-05892-0
- Rafi, H., Murray, R., Delavari, F., Perroud, N., Vuilleumier, P., Debbané, M., & Piguet, C. (2023). Neural basis of internal attention in adults with pure and comorbid ADHD. *Journal* of Attention Disorders, 27(4), 423–436. https://doi.org/10. 1177/10870547221147546
- Rifai, O. M., Fletcher-Watson, S., Jiménez-Sánchez, L., & Crompton, C. J. (2022). Investigating markers of rapport in autistic and nonautistic interactions. *Autism in Adulthood*, 4(1), 3–11. https://doi.org/10.1089/aut.2021.0017
- Scheller, M., & Sui, J. (2022). The power of the self: Anchoring information processing across contexts. *Journal of Experimental Psychology: Human Perception and Performance*, 48(9), 1001. https://doi.org/10.1037/xhp0001017
- Scheller, M., Tünnermann, J., Fredriksson, K., Fang, H., & Sui, J. (2024). Self-association enhances early attentional selection through automatic prioritization of socially salient signals. bioRxiv. https://doi.org/10.1101/2024.07.31.605951
- Skorich, D. P., & Haslam, S. A. (2022). The integrated selfcategorization model of autism. *Psychological Review*, 129(6), 1373–1393. https://doi.org/10.1037/rev0000385
- Sokol, Y., Rosensweig, C., Levin, C., & Glatt, S. (2024). Temporal self-appraisals associated with suicide-related thoughts and behaviors. *Journal of Affective Disorders*, 350, 148–154. https://doi.org/10.1016/j.jad.2024.01.081
- Stagg, S. D., Thompson-Robertson, L., & Morgan, C. (2023). Primary school children rate children with autism negatively

on looks, speech and speech content. *British Journal of Developmental Psychology*, 41(1), 37–49. https://doi.org/10. 1111/bjdp.12430

- Stanton, K., Carpenter, R. W., Nance, M., Sturgeon, T., & Villalongo Andino, M. (2022). A multisample demonstration of using the prolific platform for repeated assessment and psychometric substance use research. *Experimental and Clinical Psychopharmacology*, 30(4), 432–443. https://doi.org/10. 1037/pha0000545
- Steffens, N. K., LaRue, C. J., Haslam, C., Walter, Z. C., Cruwys, T., Munt, K. A., Jetten, J., & Tarrant, M. (2021). Social identification-building interventions to improve health: A systematic review and meta-analysis. *Health Psychology Review*, *15*(1), 85–112. https://doi.org/10.1080/17437199.2019. 1669481
- Sui, J., Greenshaw, A. J., Macrae, C. N., & Cao, B. (2021). Self research: A new pathway to precision psychiatry. *Journal of Affective Disorders*, 293, 276–278. https://doi.org/10.1016/j. jad.2021.06.041
- Sui, J., & Gu, X. (2017). Self as object: Emerging trends in self research. *Trends in Neurosciences*, 40(11) 643–653. https:// doi.org/10.1016/j.tins.2017.09.002
- Sui, J., He, X., Golubickis, M., Svensson, S. L., & Macrae, C. N. (2023). Electrophysiological correlates of self-prioritization. *Consciousness and Cognition*, 108, 103475. https://doi.org/ 10.1016/j.concog.2023.103475
- Sui, J., He, X., & Humphreys, G. W. (2012). Perceptual effects of social salience: Evidence from self-prioritization effects on perceptual matching. *Journal of Experimental Psychology: Human Perception and Performance*, 38(5), 1105–1117. https://doi.org/10.1037/a0029792
- Sui, J., & Humphreys, G. W. (2015). The integrative self: How self-reference integrates perception and memory. *Trends in Cognitive Sciences*, 19(12), 719–728. https://doi.org/10.1016/ j.tics.2015.08.015
- Sui, J., & Humphreys, G. M. (2017). Aging enhances cognitive biases to friends but not the self. *Psychonomic Bulletin & Review*, 24, 2021–2030. https://doi.org/10.3758/s13423-017-1264-1
- Sui, J., Rotshtein, P., Lu, Z., & Chechlacz, M. (2024). Causal roles of ventral and dorsal neural systems for automatic and control self-reference processing: A function lesion mapping study. *Journal of Clinical Medicine*, 13(14), 4170. https://doi.org/ 10.3390/jcm13144170
- Symons, C. S., & Johnson, B. T. (1997). The self-reference effect in memory: A meta-analysis. *Psychological Bulletin*, *121*(3), 371–394. https://doi.org/10.1037/0033-2909.121.3. 371
- Turnock, A., Langley, K., & Jones, C. R. (2022). Understanding stigma in autism: A narrative review and theoretical model. *Autism in Adulthood*, 4(1), 76–91. https://doi.org/10.1089/aut. 2021.0005
- Verplanken, B., & Sui, J. (2019). Habit and identity: Behavioral, cognitive, affective, and motivational facets of an integrated self. *Frontiers in Psychology*, 10, 1504. https://doi.org/10. 3389/fpsyg.2019.01504

- Visser, M. J., Peters, R. M., & Luman, M. (2024). Understanding ADHD-related stigma: A gender analysis of young adults and key stakeholder perspectives. *Neurodiversity*, 2, 2754633 0241274664. https://doi.org/10.1177/27546330241274664
- Watts, G., Crompton, C., Grainger, C., Long, J., Botha, M., Somerville, M., & Cage, E. (2024). 'A certain magic'–autistic adults' experiences of interacting with other autistic people and its relation to quality of life: A systematic review and thematic meta-synthesis. *Autism*, 13623613241255811. https://doi.org/ 10.1177/13623613241255811
- Whelpley, C. E., & May, C. P. (2023). Seeing is disliking: Evidence of bias against individuals with autism spectrum disorder in traditional job interviews. *Journal of Autism and Developmental Disorders*, 53(4), 1363–1374. https://doi.org/ 10.1007/s10803-022-05432-2
- Williams, D. M., Nicholson, T., & Grainger, C. (2018). The selfreference effect on perception: Undiminished in adults with autism and no relation to autism traits. *Autism Research*, 11(2), 331–341. https://doi.org/10.1002/aur.1891

- Wuyun, G., Wang, J., Zhang, L., Yi, L., & Wu, Y. (2020). Actions speak louder than words: The role of action in self-referential advantage in children with autism. *Autism Research*, 13(5), 810–820. https://doi.org/10.1002/aur.2274
- Yankouskaya, A., Lovett, G., & Sui, J. (2023). The relationship between self, value-based reward, and emotion prioritisation effects. *Quarterly Journal of Experimental Psychology*, 76(4), 942–960. https://doi.org/10.1177/1747021822110 2887
- Yankouskaya, A., & Sui, J. (2022). Self-prioritization is supported by interactions between large-scale brain networks. *European Journal of Neuroscience*, 55(5), 1244–1261. https://doi.org/ 10.1111/ejn.15612
- Zhuang, S., Tan, D. W., Reddrop, S., Dean, L., Maybery, M., & Magiati, I. (2023). Psychosocial factors associated with camouflaging in autistic people and its relationship with mental health and well-being: A mixed methods systematic review. *Clinical Psychology Review*, 105, 102335. https:// doi.org/10.1016/j.cpr.2023.102335